

Advancing Treatment Strategies in Linitis Plastica. Role of Surgery

Reviewer 1

Answer to Comments:

- 1: We agreed that goes against common sense and deleted the sentence.
- 2: We reviewed the text and added the following: “Xu et. al. Xu et. al. conducted a study to... LP patients with type 3 Gastric Cancer”
- 3: The title has changed to “Therapeutic Strategies and Prognostic Challenges in Linitis Plastica”
- 4: We added “Linitis Plastica is a characteristic... locally advanced stages.” And “Metastatic linitis plastica... scirrhous carcinoma of the stomach.”
- 5: We added two paragraphs “As regards the impact of surgery...plus adjuvant chemotherapy” and “The results from these... these patient groups.”
- 6: We have added “A study conducted by Chengcai L. et. al.... role of surgery is quite limited. (24)”
- 7: We have added “Given the complexities... of its management strategy.”, “Despite the central role... other treatment modalities.” And “As regards to the strategies,... in the area of treatment.”
- 8: We corrected “2, 6, 7, 11, 12, 13, 14, 15, 16, 18”

Reviewer 2

Answer to Comments:

1. We have added “Linitis Plastica is a characteristic finding of scirrhous gastric carcinoma... locally advanced stages [7]” that supplements the rest text.

2. This comment includes a valuable suggestion. While we acknowledge the importance of a detailed discussion on the controversies surrounding treatment strategies, we believe that the current scope and focus of the editorial provide a balanced overview. Including an in depth exploration of differing expert opinions might shift the focus away from the primary objectives of our study.
3. We agree that patient compliance and quality of life are essential aspects of treatment planning but our editorial primarily focuses on clinical outcomes and treatment efficacy.
4. We added “Future studies should focus...each patient hat.”